Author’s response to reviews

Title: Coix lacryma-jobi var. ma-yuen Stapf sprout extract has anti-metastatic activity in colon cancer cells in vitro

Authors:

Eun Suk Son (eunsuk0607@hanmail.net)
Young Ock Kim (kyo9128@korea.kr)
Chun Geon Park (pcg@korea.kr)
Kyung Hun Park (blueour@korea.kr)
Sung Hwan Jeong (jsw@gilhospital.com)
Jeong Woong Park (jwpark@gilhospital.com)
Se-Hee Kim (sehee0423@gilhospital.com)

Version: 2 Date: 15 Oct 2017

Author’s response to reviews:

1. Please include the email addresses of all authors on the title page of your manuscript.

Response: We included the email addresses of all authors on the title page of the manuscript as follows:

Authors details

1Department of Medicine, Gachon University Gil Medical Center, Incheon, Republic of Korea; eunsuk0607@hanmail.net (E.S.S); jsw@gilhospital.com (S.H.J.); jwpark@gilhospital.com (J.W.P.)

2Department of Herbal Crop Research, National Institute of Horticultural and Herbal Science, RDA, Chungbuk, Republic of Korea; kyo9128@korea.kr (Y.O.K.); pcg@korea.kr (C.G.P.); blueour@korea.kr (K.H.P.)

3Gachon Medical Research Institute, Gachon University Gil Medical Center, Incheon, Republic of Korea; sehee0423@gilhospital.com (S.-H.K.)
2. Thank you for providing a section on author contribution. We use this information to note which author contributed what to the study. We had noticed only six of the seven authors are mentioned in this section. Please revise this section to include all authors, so that all authors listed in the manuscript can qualify for authorship.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. We respectfully request you please amend this statement to adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. According to the ICMJE guidelines, to qualify as an author one should have:

- made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

- been involved in drafting the manuscript or revising it critically for important intellectual content;

- given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and

- agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship.

If you would please amend the statement to reflex these requirements, we would appreciate it.

Additionally, please use this statement to confirm that all authors have read and approved of the final manuscript, and all authors are referred to by their initials.

Response: All seven authors listed in the manuscript had already been mentioned in the authors’ contributions as follows:

Authors’ contributions

Eun Suk Son and Young Ock Kim equally contributed to the research; Chun Geon Park and Kyung Hun Park provided Coix lacryma-jobi var. ma-yuen Stapf sprout extract for this research; Sung Hwan Jeong analyzed the data; Jeong Woong Park and Se-Hee Kim wrote the manuscript and supervised this work.

3. Please upload all of your figures individually on the submission system.
Response: We will have uploaded all the figures individually via the submission system by the time you read this.